Author's response to reviews

Title: Efficacy of sorafenib on metastatic renal cell carcinoma in Asian patients: Results from a multicenter study

Authors:

Hailiang Zhang (zhanghl918918@yahoo.com.cn)
Baijun Dong (dongbajun@hotmail.com)
Jiade J Lu (sintelligence@gmail.com)
Xudong Yao (xudongyao67@shca.org.cn)
Shilin Zhang (zhangsl1969@shca.org.cn)
Bo Dai (bodai1978@shca.org.cn)
Yijun Shen (yijunshen@shca.org.cn)
Yao Zhu (zhu_yao@shca.org.cn)
Dingwei Ye (dwye@shca.org.cn)
Yiran Huang (hyrrenji2@yahoo.com.cn)

Version: 8 Date: 20 July 2009

Author's response to reviews:

The manuscript was edited according to the requirement previously and was accepted. This is the proof-read final version of our manuscript. Some typo and minor mistakes including the title of the hospital (cancer center instead of cancer hospital) was corrected. Many thanks for accepting our manuscript for publishing.

Best regards,
Professors Ye and Huang